Author’s response to reviews

Title: Postnatal Care Utilization and Associated Factors Among Women of Reproductive Age Group in Halaba Kulito Town, Southern Ethiopia

Authors:
teshome Abebo (teshabuka@gmail.com)
Dawit Tesfaye (devanhijember@gmail.com)

Version: 5 Date: 16 Nov 2017

Author’s response to reviews:

Point by point response

Interactions were checked for variables found at the final model in multivariable analysis. Among interaction terms created, only knew child care by frequency of ANC had significant effect with PNC services utilization. Hence, interaction term, knew child care by frequency of ANC, was included in the final model.

Also we made revision to result, discussion and conclusion parts according to results obtained from model containing interaction

We provided two tables as Appendix I and Appendix II which show result of multicollinearity (Appendix I) and model with interaction terms (Appendix II)

Also we made additional copy editing revision